Author's response to reviews

Title: Fusion beat in patients with heart failure treated with Left Ventricular Pacing: may ECG morphology relate to mechanical synchrony? A pilot study

Authors:

Lorella Gianfranchi (l.gianfranchi@ausl.fe.it)
Katia Bettiol (laboratorioecocento@ausl.fe.it)
Biagio Sassone (biagio.sassone@ausl.bo.it)
Roberto Verlato (roberto.verlato@libero.it)
Giorgio Corbucci (giorgio.corbucci@vitatron.com)
Paolo Alboni (p.alboni@ausl.fe.it)

Version: 4 Date: 23 December 2007

Author's response to reviews: see over
Dear Editor

The resubmitted version of the paper was written in accordance with the requested formatting changes listed below:

1. Title, author list on one line, affiliations written in full, in the following format: department, institute, city, and country linked to each author via superscript numbers, corresponding author and then finally authors' e-mail addresses.
2. Abstract is present in the manuscript succeeding the title page and matches exactly what is in the submission system.
3. Authors' contributions were reported in accordance with the format of the journal.
4. The corresponding author is indicated on the title page.
5. Reference list now contains all authors for each publication.
6. The terms “Figure legends” replaces “Legends”.
7. Table 1 was created by Table Autoformat function in MS Word. Its legend appears below the table.

Best Regards
Lorella Gianfranchi